Author's response to reviews

Title: Deregulation of microRNAs Let-7a and miR-21 mediate aberrant STAT3 signaling during Human Papillomavirus-induced Cervical Carcinogenesis: Role of E6 Oncoprotein

Authors:

Gauri Shishodia (gaurishishodia@gmail.com)
Gaurav Verma (gverma.gaurav@gmail.com)
Yogesh Srivastava (yansh111@gmail.com)
Ravi Mehrotra (rm8509@gmail.com)
Bhudev C. Das (bcdas48@hotmail.com)
Alok C. Bharti (bhartiac@icmr.org.in)

Version: 6 Date: 26 November 2014

Author's response to reviews:

Dear BioMed Central Team
As requested the abbreviations in acknowledgement section and grants section have been removed and expended forms have been mentioned.
Best wishes
AC Bharti